Author’s response to reviews

Title: Microorganisms involved in Deep Neck Infection (DNIs) in Greece: detection, identification and susceptibility to antimicrobials

Authors:

Despoina Beka (beka@gmail.com)
Vasileios Lachanas (lachanas@gmail.com)
Stergios Doumas (doumas@gmail.com)
Stelios Xytsas (xytsas@gmail.com)
Anastasios Kanatas (kanatas@gmail.com)
Efi Petinaki (petinaki@med.uth.gr)
Charalampos Skoulakis (skulakis@hotmail.com)

Version: 3 Date: 12 Sep 2019

Author’s response to reviews:

Dear Editor,

Thank you very much for your decision concerning our manuscript entitled “Microorganisms involved in Deep Neck Infection (DNIs) in Greece: detection, identification and susceptibility to antimicrobials”. In this revised version-R3 the corrections suggested were done.

Sincerely yours,

Efi Petinaki, MD
Professor of Clinical Microbiology,
Medical School, University of Larissa,
Greece